Author's response to reviews

Title: Activation of endogenous p53 by combined p19Arf gene transfer and nutlin-3 drug treatment modalities in the murine cell lines B16 and C6

Authors:

Christian A. Merkel (cmerkel@usp.br)
Rafael B. da Silva Soares (rafaelbss@yahoo.com.br)
Anna Carolina V. de Carvalho (carol_tite@yahoo.com.br)
Daniela B. Zanatta (danizanatta@gmail.com)
Marcio C. Bajgelman (mcbajgelman@yahoo.com.br)
Paula Fratini (fratini@usp.br)
Eugenia Costanzi-Strauss (ecostanz@usp.br)
Bryan E. Strauss (bstrauss@usp.br)

Version: 3 Date: 9 June 2010

Author's response to reviews:

The manuscript has been revised as directed by the editorial staff. Specifically, the author names follow the format suggest and match the author list provided online. Highlighting has been removed. Author affiliations contain the country. The running title and key words have been removed. Author email addresses have been provided. The header has been removed.